Author's response to reviews

Title: Incidence of and risk factors for Motor Neurone Disease in UK women: A prospective study

Authors:

   Pat Doyle (pat.doyle@lshtm.ac.uk)
   Anna Brown (anna.brown@ceu.ox.ac.uk)
   Valerie Beral (pa.valerie.beral@ceu.ox.ac.uk)
   Gillian Reeves (gillian.reeves@ceu.ox.ac.uk)
   Jane Green (jane.green@ceu.ox.ac.uk)

Version: 3 Date: 7 February 2012

Author's response to reviews:

This revision contains two small edits made in response to an email of 7th February form Gerard Clunn of the BMC Editorial Team. (i) Deletion of the words "and the Million women Study collaborators" from the list of authors, and (ii) the addition of an Acknowledgement section at the end of the text, which refers to details of the Million women Study collaborators.